Author’s response to reviews

Title: Induction of the interleukin 6/ signal transducer and activator of transcription pathway in the lungs of mice sub-chronically exposed to mainstream tobacco smoke.

Authors:

Sabina Halappanavar (sabina_halappanavar@hc-sc.gc.ca)
Marsha Russell (marsha_russell@hc-sc.gc.ca)
Martin R Stampfli (stampfli@mcmaster.ca)
Andrew Williams (Andrew_williams@hc-sc.gc.ca)
Carole L Yauk (carole_yauk@hc-sc.gc.ca)

Version: 2 Date: 25 November 2008

Author’s response to reviews:

As instructed the following changes have been made to the main body of the manuscript.
1. abstract: It has been rearranged into background, methods, results and conclusions.
2. Manuscript section is structured in the following order: abstract, background, methods, results, discussion, conclusion, competing interests, author’s contribution, acknowledgements, references, figure legends
3. Competing interest: A declaration is made answering all the questions.